Author's response to reviews

Title: Managing symptoms during cancer treatments: Evaluating the implementation of evidence-informed remote support protocols

Authors:

Dawn Stacey (dstacey@uottawa.ca)
Debra Bakker (dbakker@laurentian.ca)
Barbara Ballantyne (bballantyne@hsnsudbury.ca)
Kimberly Chapman (kim.chapman@horizonnb.ca)
Joanne Cumminger (joanne.cumminger@pcha.nshealth.ca)
Esther Green (esther.green@cancercare.on.ca)
Margaret Harrison (margaret.b.harrison@queensu.ca)
Doris Howell (doris.howell@uhn.on.ca)
Craig Kuziemsky (kuziemsky@telfer.uottawa.ca)
Terry MacKenzie (tmackenzie@hsnsudbury.on.ca)
Brenda Sabo (brenda.sabo@dal.ca)
Myriam Skrutkowski (mskathome@gmail.com)
Ann Syme (enzyme@shaw.ca)
Angela Whynot (angela.whynot@cdha.nshealth.ca)

Version: 6 Date: 6 November 2012

Author's response to reviews:

Thank you for the opportunity to review the copy-edited version of our manuscript. We have accepted all the suggested changes.

Dawn Stacey RN, PhD
Associate Professor, University of Ottawa
Scientist, Ottawa Hospital Research Institute